Author's response to reviews

Title: Harpalyacin 2 inhibits the enzymatic and platelet aggregation activities of PrTX-III, a D49 phospholipase A2 from Bothrops pirajai venom.

Authors:

Rafael M Ximenes (ximenesrm@gmail.com)
Renata S Alves (renatasalves@gmail.com)
Renata M Araújo (renat.onca@gmail.com)
Edilberto R Silveira (edil@ufc.br)
Marcelo M Rabello (montenegro.rabello@gmail.com)
Marcelo Z Hernandes (zaldini@ufpe.br)
Verônica CG Soares (vcsaures@gmail.com)
Daniel Bristot (dbristot@hotmail.com)
Camila L Pires (lehnhardtipes@gmail.com)
Daniela O Toyama (daniela.toyama@mackenzie.br)
Henrique H Gaeta (henriquehg@gmail.com)
Helena SA Monteiro (hsazul@gmail.com)
Marcos H Toyama (mhtjpn@yahoo.com)

Version: 2 Date: 20 April 2012

Author's response to reviews: see over
Harpalycin 2 inhibits the enzymatic and platelet aggregation activities of PrTX-III, a D49 phospholipase A$_2$ from *Bothrops pirajai* venom.

Dear Editor,

The Ethical Statement was placed in the subsection (Platelet aggregation studies) of the Material and Methods that describes the use of human material. All experiments using human material were carried out according to Helsinki Declaration and were approved by the Ethical Committee for Human Research of State University of Campinas under the no. 0323.0.1.146.000-09.

We apologized for the delay in the revision process.

Kind regards,

Marcos H. Toyama

**Corresponding Author.** Prof. Dr. Marcos Hikari Toyama. Praça Infante Dom Henrique, s/n, Zip code: 11330-900 São Vicente, SP, Brazil. Phone: 55 13 34699400. Fax: 55 13 35699446. Email: mhtjpn@yahoo.com